Author’s response to reviews

Title: Web-based guided self-help for employees with depressive symptoms (Happy@Work): design of a randomized controlled trial

Authors:

Anna S Geraedts (a.s.geraedts@vu.nl)
Annet M Kleiboer (a.m.kleiboer@vu.nl)
Noortje M Wiezer (noortje.wiezer@tno.nl)
Willem van Mechelen (w.vanmechelen@vumc.nl)
Pim Cuijpers (p.cuijpers@vu.nl)

Version: 3 Date: 4 April 2012

Author’s response to reviews: see over
To: BMC Psychiatry

Date: 4 April 2012

Dear Editor,

This cover letter accompanies a manuscript submitted to the journal ‘BMC Psychiatry’. The manuscript is entitled ‘Web-based guided self-help for employees with depressive symptoms (Happy@Work): design of a randomized controlled trial ’. The study protocol was approved by the ethics committee of the VU University Medical Center Amsterdam, the Netherlands. These ethical standards are in accordance with the APA ethical standards. All of the authors have contributed significantly to the manuscript.

The manuscript contains the study protocol on the effect of a web-based intervention course called Happy@Work. The intervention course specifically focus on employees with depressive symptoms. Psychosocial problems, especially depressive symptoms, are very common among the working population. This leads to excessive cost for society and companies. Not only can this course, if proven effective, be implemented in occupational healthcare, but also in mental healthcare and general healthcare.

The goal of this course is to prevent depression. Early treatment of mild depressive symptoms may prevent the worsening of depressive symptoms into a depression. This goal matches the scopes of BMC Psychiatry and may be of interest for healthcare professionals and healthcare researchers.

I have made the following changes after your email from March 8th:
- An email is sent to editorial@biomedcentral.com with evidence of ethical approval and external non commercial funding (both in English).
- This study is reviewed and funded by the EMGO+ Institute for Health and Care research (see funding letter).
- In the manuscript I have made comments on related articles as requested.

Yours sincerely,
Anna Geraedts
PhD-student
Vrije University Amsterdam
Van der Boechorststraat 1, 1081 BT Amsterdam
The Netherlands
Phone: +31 20 5987451/ Fax: +31 20 5988758
Email: a.s.geraedts@vu.nl